Author’s response to reviews

Title: Mental health problems in the 10th grade and non-completion of upper secondary school: the mediating role of grades in a population-based longitudinal study

Authors:

Åse Sagatun (aase.sagatun@r-bup.no)
Sonja Heyerdahl (sonja.heyerdahl@r-bup.no)
Tore Wentzel-Larsen (tore.wentzel-larsen@r-bup.no)
Lars Lien (lars.lien@medisin.uio.no)

Version: 3 Date: 22 November 2013

Author’s response to reviews:

Dear Editor

Thank you for valuable comments to our manuscript “Mental health problems in the 10th grade and non-completion of upper secondary school: the mediating role of grades in a population-based longitudinal study”.

Attached is a point-by-point response to the concerns.

Best regards, Åse Sagatun